Author's response to reviews

Title: Pragmatic randomised controlled trial of group psychoeducation versus group support in the maintenance of bipolar disorder.

Authors:

Richard K Morriss (richard.morriss@nottingham.ac.uk)
Fiona Lobban (f.lobban@lancaster.ac.uk)
Steven Jones (S.jones7@lancaster.ac.uk)
Lisa Riste (lisa.riste@manchester.ac.uk)
Sarah Peters (sarah.peters@manchester.ac.uk)
Christopher Roberts (chris.roberts@manchester.ac.uk)
Linda Davies (linda.davies@manchester.ac.uk)
Debbie Mayes (d.mayes@lancaster.ac.uk)

Version: 2  Date: 22 June 2011

Author's response to reviews:

School of Community Health Sciences
Division of Psychiatry
B Floor
The Institute of Mental Health
The Sir Colin Campbell Building
University of Nottingham Innovation Park
Triumph Road
Nottingham
NG7 2TU

Tel +44 (0)115 823 0427
Fax +44 (0)115 823 1289

22nd June 2011

Sabina Alam, PhD
Senior Executive Editor
BMC series journals
Email: richard.morriss@nottingham.ac.uk

Professor of Psychiatry and Community Health

Dear Sabina
RE: MS: 1964995397547994 Pragmatic randomised controlled trial of group psychoeducation versus group support in the maintenance of bipolar disorder.

Thank you for accepting the above paper subject to the addition of some important information.

1. ISRCTN number. Sorry the ISRCTN number submitted should be ISRCTN62761948 not ISRCTN627461948.
2. A statement has been added concerning conflict of interest.
3. A statement has been made concerning authorship of the paper.

In addition in the acknowledgements I have included a form of words that our funders wish to be used in the paper to the effect that they funded the study but the views expressed are those of the authors rather than the funders or NHS.

I hope the paper is now suitable for publication but I would be glad to assist you further with it if there are any further changes that are needed.

Yours sincerely

Richard Morriss
Professor of Psychiatry and Community Mental Health